Author's response to reviews

Title: Evaluation of procalcitonin for diagnosis of neonatal sepsis of vertical transmission

Authors:

Jose B Lopez Sastre (jose_lopez_sastre@hotmail.com)
David Perez Solis (david.perez@sespa.princast.es)
Vicente Roques Serradilla (roes5@ono.com)
Belen Fernandez Colomer (bcolomer@arrakis.es)
Gil D Coto Cotallo (gildaniel.coto@sespa.princast.es)
Grupo de Hospitales Castrillo (ghc) (gcastrillo@arrakis.es)

Version: 6 Date: 25 February 2007

Author's response to reviews:

Dear members of The BioMed Central Editorial Production Team,

We are delighted you have finally accepted to publish our manuscript. We have submitted a new version of the manuscript, including format changes you have requested.

- The category has been removed from above the title.
- Author groups: we have included the author group in the author list, providing a full affiliation, both in manuscript and in the submission system. We have included the first five authors, as coordinators of the study, in the title page and submission system, and the rest of the listed members are now in the Acknowledgements section defining.
- We have included a Conclusions section after the Discussions.
- Acknowledgements: we describe the sources of funding for the study and for the manuscript preparation. The funders had no role in study design, analysis of data, writing of the manuscript, or in the decision to submit the manuscript for publication.

Sincerely,

David Perez Solis, MD, PhD
Jose Blas Lopez Sastre, MD, PhD
Hospital Universitario Central de Asturias, Spain